

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Albers, Ronald; Vijselaar, Focco

Working Paper New technologies and productivity growth in the euro area

ECB Working Paper, No. 122

Provided in Cooperation with: European Central Bank (ECB)

Suggested Citation: Albers, Ronald; Vijselaar, Focco (2002) : New technologies and productivity growth in the euro area, ECB Working Paper, No. 122, European Central Bank (ECB), Frankfurt a. M.

This Version is available at: https://hdl.handle.net/10419/152556

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

EUROPEAN CENTRAL BANK

WORKING PAPER SERIES



WORKING PAPER NO. 122

NEW TECHNOLOGIES AND PRODUCTIVITY GROWTH IN THE EURO AREA

BY FOCCO VIJSELAAR AND RONALD ALBERS

February 2002

EUROPEAN CENTRAL BANK

WORKING PAPER SERIES



WORKING PAPER NO. 122

NEW TECHNOLOGIES AND PRODUCTIVITY GROWTH IN THE EURO AREA

BY FOCCO VIJSELAAR AND RONALD ALBERS'

February 2002

I Economists, Directorate Economic Developments, European Central Bank. We thank Bart van Ark, Neale Kennedy, Gerard Korteweg, Ad van Riet, Marcel Timmer and two anonymous referees for their comments. All errors and omissions remain ours, of course. We thank Erikos Velissaratos for his help in acquiring data on investment in ICT and Colin Webb for providing us with the OECD STAN database. This paper represents the views of the authors and does not necessarily reflect the views of the European Central Bank.

© European Central Bank, 2002

Address	Kaiserstrasse 29
	D-60311 Frankfurt am Main
	Germany
Postal address	Postfach 6 03 9
	D-60066 Frankfurt am Main
	Germany
Telephone	+49 69 344 0
Internet	http://www.ecb.int
Fax	+49 69 1344 6000
Telex	411 144 ecb d

All rights reserved.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged. The views expressed in this paper are those of the authors and do not necessarily reflect those of the European Central Bank.

ISSN 1561-0810

Contents

Abstr	ract	4
Non-	technical summary	5
I	Introduction	7
2	Related literature	9
3	Sectoral developments	П
4	Contribution of new technologies to economic growth	16
5	Concluding remarks	22
Refer	rences	23
Арре	ndix I Data sources and aggregation methods	26
Арре	ndix 2 Sectoral developments in individual euro area countries	29
Арре	ndix 3 Growth accounting for individual countries	31
Арре	ndix 4 Measurement problems related to productivity	35
	for ICT goods	35
	4.2 Other measurement issues	38
Europ	bean Central Bank Working Paper Series	40

Abstract

This paper provides an overview of the currently available evidence on the importance of information and communication technologies (ICT) for developments in productivity growth in the euro area. On the basis of the available data, there is evidence of an increased contribution of ICT to economic growth both in terms of production and investment in the second half of the 1990s. However, there is little, if any, evidence of significant positive spillover effects from the use of ICT to overall productivity growth. This implies that there is no reason to believe that potential output growth in the euro area has increased significantly in recent years on account of new technologies.

JEL classification: E22, L63, L86, O3, O47

Key words: Information and communication technologies, average labour productivity, sectoral developments, growth accounting, capital stock, euro area, measurement issues

Non-technical summary

In the period from the mid-1990s to 2000, the macroeconomic performance of the United States was remarkable. For instance, over this period, average labour productivity (ALP) growth - i.e. the increase in output per person employed or, preferably, the increase in output per hour worked - for instance, clearly increased, while employment continued to grow at a steady rate. In explaining this performance, the focus of recent research has mainly been on the revolution in information and communication technologies (ICT).

An important question in this respect is whether these new technologies have also had an impact on productivity growth in the euro area. In contrast to the United States, there is only scant evidence for the euro area of the impact of the new technologies on economic developments. The aim of this paper is to partly redress the balance by studying the importance of ICT for productivity growth in the euro area. To our knowledge, this is the first contribution that focuses on the euro area as such.

Growth in ALP can be the result of an increase in the amount of capital available per hour worked (capital deepening) or of an increase in the overall efficiency of the economic process, as measured by the gain in total factor productivity (TFP). In other words, TFP growth can be interpreted as that part of overall productivity growth that cannot be accounted for by higher capital or labour input. ICT could lead to higher ALP growth trough capital deepening and, if the use of ICT improves the efficiency of the economic process, also through an increase in TFP growth. However, it is difficult to disentangle the forces driving TFP growth. As opposed to ALP, TFP growth cannot be measured directly and is difficult to estimate in practice.

This paper follows two approaches to gauge the importance of ICT for euro area productivity growth. First, it directly accounts for the contribution of ICT capital to ALP growth and estimates developments in TFP growth in the euro area by applying a standard growth accounting framework, which decomposes the sources of output growth. Second, it examines in more detail the developments in ALP growth at the sectoral level, focusing on sectors producing and intensively using ICT.

The main findings may be briefly summarised as follows.

The results of the growth accounting exercise suggest that the importance of ICT capital accumulation for economic growth in the euro area has increased in the second half of the 1990s. The euro area is thus experiencing positive growth effects of ICT through capital deepening. However, TFP growth has been declining rather than accelerating in the course of the last decade. This casts doubt on a significant positive impact of the use of ICT on the increase in efficiency of the economic process in the euro area.

Output and ALP growth in ICT *producing* sectors in the euro area have clearly been higher than in other sectors of the economy. This points to a positive impact of ICT on economic growth. However, the size of these sectors is relatively small implying that there has so far been only a limited impact on overall economic developments. Nevertheless, the aggregate contribution of ICT sectors to total ALP growth was noticeable in the euro area in the period 1991-1998. The fact that the growth rates of ALP in the ICT *using* sectors did not rise appreciably faster than in the non-ICT using sectors again casts doubt on the existence of a positive impact from the use of ICT on TFP growth.

The conclusion can thus be that ICT was of increasing importance for economic growth in the euro area over the 1990s. At the same time, there is no reason to believe that the significant rise in ICT investment in the course of the last decade has led to a significant rise in economy-wide TFP growth. These findings do also not support the notion that ICT has raised potential output growth in the euro area.

Finally, a further interesting result is that the contribution of ICT capital to output growth has not been much different in the euro area than in the United States, if one attempts to allow for the effects of different deflation techniques – contrary to conventional wisdom. This suggests that other factors are likely to account for the largest part of the observed difference in TFP growth, including differences in production structure and possibly also in the flexibility of product, labour, and financial markets.

I Introduction

In the period from the mid-1990s to 2000, the macroeconomic performance of the United States was remarkable. Over this period, average labour productivity (ALP) growth in the United States, for instance, clearly increased, while employment continued to grow at a steady rate. In explaining this performance, the focus of recent research has mainly been on the revolution in information and communication technologies (ICT). In some studies it is argued that it is mainly the ICT producing sector that has been responsible for the increase in productivity growth (Jorgenson and Stiroh 2000; Jorgenson 2001; Stiroh 2001a, 2001b), while others argue that in addition the use of ICT goods and services has contributed to the acceleration in productivity (Oliner and Sichel 2000; Bailey and Lawrence 2001; Nordhaus 2001).²

An important question in this respect is whether these new technologies have also had an impact on productivity growth in the euro area. In contrast to the United States, there is only scarce evidence for the euro area of the impact of the new technologies on economic developments. Thus, even though in recent years the (policy) debate on the impact of ICT on productivity growth has raged on both sides of the Atlantic Ocean, to date the vast majority of empirical studies has remained limited to the United States. The aim of this paper is to partly redress the balance by studying the importance of ICT for productivity growth in the euro area. To our knowledge, this is the first paper that focuses on the euro area as such.

Moreover, the present paper has a broader scope than the relatively few empirical studies on individual euro area countries. First, this study presents evidence on sectoral developments in ALP. Taking this perspective, it is important to distinguish increases in productivity growth resulting from developments limited to the ICT producing sectors from increases due to the spreading use of ICT in other sectors of the economy. Arguably, only if ICT has the character of a so-called general-purpose technology would it result in a more rapid sustained increase in the overall efficiency of the economic process, which would imply that the economy has a higher rate of growth of potential output.³ Second, this paper focuses on a decomposition of the sources of productivity growth rather than limiting itself to an estimate on the contribution of ICT capital to output growth as is done

 $^{^2}$ The alleged revolutionary character of ICT has not been undisputed. Gordon (2000), for instance, remains sceptical about the importance of ICT, arguing that it does not measure up to the great inventions of the past – in particular electricity and the internal combustion engine - in affecting productivity and the quality of life. ³ Note however, that there is no clear-cut definition of what are the defining elements of a general-purpose

³ Note however, that there is no clear-cut definition of what are the defining elements of a general-purpose technology. The historical experience shows that the impact of major technological breakthroughs on macroeconomic productivity developments has not been comparable across technologies, which makes any comparison with previous episodes hazardous (Wellink and Albers 2001). Furthermore, in theory it would be sufficient for ICT production and investment to increase at a sufficiently high rate to raise potential output and increase in the growth rate of output, without necessarily being spread widely over the economy

in most studies on individual euro area countries. This approach does allow for making inferences about potential spillover effects of the use of ICT.

Productivity growth is most often measured in terms of ALP, i.e. as the increase in output per person employed or, preferably, the increase in output per hour worked. Growth in ALP can be the result of an increase in the amount of capital available per hour worked (capital deepening) or of an increase in the overall efficiency of the economic process, as measured by the gain in total factor productivity (TFP). An acceleration in ALP due to an increase in TFP growth could be a sign of the general-purpose character of ICT. However, as opposed to ALP, TFP growth cannot be measured directly and is difficult to estimate in practice.

As mentioned, this paper uses two approaches to gauge the importance of ICT for euro area productivity growth. First, it directly accounts for the contribution of ICT capital to ALP growth and estimates developments in TFP growth in the euro area by applying a standard growth accounting framework. This framework has been widely applied in studies on aggregate data on the United States, in particular. However, most recently the emphasis in the analysis of economic effects of ICT has shifted from aggregate-level to industry-level studies (e.g. Stiroh 2001a, McKinsey Global Institute 2001, Van Ark 2000 and 2001). As data on ICT investment by industry is still largely unavailable for euro area countries, it is not possible to undertake a sectoral growth accounting exercise. Therefore, a second focal point will be on developments in ALP growth in sectors producing and intensively using ICT. Due to data constraints this study largely limits itself to developments in the 1990s. While the emphasis is clearly on the euro area, comparisons with the United States are drawn to add a comparative perspective.

The remainder of the paper is structured as follows. Section 2 discusses the relevant literature on the importance of ICT for economic growth in euro area countries. Section 3 presents the results of the analysis on sectoral developments, while section 4 presents the results of the growth accounting exercise. Section 5 concludes. The appendices provide details on data sources and estimation methods, on individual country results, and on the likely sources of measurement error and the overall impact of using alternative (US based) deflators for IT equipment to the euro area estimates.

2 Related literature

So far, only a few studies have considered the impact of ICT on economic growth in euro area countries. This is mainly due to a relative dearth of national accounts data on investment in ICT. Most studies have therefore used private sector data sources on ICT expenditure to construct investment series for euro area countries. These have the disadvantage that they are not consistent with national accounts methodologies and thus are not directly comparable to the available official statistics. This study, however, uses official data on both the production of and investment in ICT.

Turning to the related literature, Schreyer (2000) used data for G7 countries from a private data source (International Data Corporation) on ICT expenditure and computed the contribution to growth of ICT capital by applying a standard growth accounting framework. He estimated that ICT capital contributed some 0.4 percentage point per annum on average to economic growth in the United States in the period 1990-1996, compared to about 0.2 percentage point in Germany, France and Italy over the same period. Schreyer made no estimates of TFP growth.

Daveri (2001) and Colecchia and Schreyer (2001) extended the work of Schreyer (2000) and updated the estimates to 1999. Daveri's study covered all EU countries (with the exception of Luxembourg) as well as the United States. An important extension is the incorporation of investment in software. This, together with a different method to construct investment series from the expenditure data, led to higher estimates of investment in ICT in EU countries. With software included the contribution of ICT capital to real GDP growth in EU countries varied from 0.3 to 0.6 percentage point in the period 1991-1999, compared to 0.9 percentage point in the United States over the same period. In most countries, the contribution to growth in real business sector output was found to have increased from the first to the second half of the 1990s. Daveri also estimated TFP growth, which he found to have increased in five smaller euro area countries only.

Colecchia and Schreyer (2001) derived estimates of contributions of ICT capital to output growth in the business sector on the basis of official data for three euro area countries (France, Germany, and Italy). In addition, they constructed some estimates for Finland. They estimated the contribution of ICT to business output growth to vary from 0.2 to 0.3 percentage point using national deflators and from 0.3 to 0.6 percentage point using alternative US based deflators in the period 1995-1999. With the notable exception of Germany, they found a clear increase in the contribution of ICT to output growth from the first to the second half of the 1990s. The study did not provide estimates of TFP growth.

In appendix 2 the results of Colecchia and Schreyer are presented in more detail and compared with the results of this study.

Using another private sector data source on ICT expenditure (Reeds), Roeger (2001) presented different scenarios for the contribution of ICT capital to output growth in EU countries. In particular he used various assumptions as regards the price deflators for ICT goods and the price elasticity of ICT capital. The estimated contributions of ICT to output growth varied from 0.2 to 0.3 percentage point in the years 1992-1994, and from 0.3 to 0.6 percentage point in the period 1995-1999 – with an increase in the contribution in each case between the earlier and the later period. Moreover, in estimating the effect of ICT on aggregate TFP growth, Roeger concluded that there was little evidence of substantial spillovers on account of the use of ICT.

Three further papers draw on national accounts data to make bilateral comparisons with the United States. Mairesse et al. (2000) compared France with the United States. They found that for France the contribution of ICT (including software) to output growth increased from 0.2 percentage point in the period 1989-1995 to 0.3 percentage point in the period 1995-1999. This contribution has continued to grow in recent years, to reach 0.4% in 1999. Two papers (CPB 2000, Van der Wiel 2001) focused on the contribution of ICT to aggregate and sectoral ALP growth in the Netherlands. CPB (2000) found that ICT (excluding software) contributed 0.2 percentage point to growth in ALP in both the periods 1991-1995 and 1996-1999. Van der Wiel (2001) presented additional estimates of the contribution of software to ALP growth in the Netherlands, which increased from 0.1 percentage point in the first period to 0.2 in the second. In a study on Finland, Jalava and Pohjola (2001) concluded, using data on ICT expenditure rather than national accounts data, that the contribution of ICT capital to output growth in Finland increased from 0.3 percentage points in the early 1990s to 0.7 percentage point in the late 1990s, mainly on account of investment in communication equipment.

A paper by Van Ark (2001) differs from the studies mentioned above in that it highlights the production side rather than the investment side of ICT, by analysing developments in ICT producing and ICT using sectors in ten major OECD countries. The main finding of Van Ark was that the differences in ALP growth between the United States and most European countries are partly explained by a larger and more productive ICT producing sector in the United States, and also by a higher contribution to productivity in the United States from ICT using industries and services. This notwithstanding, in general ALP growth in the ICT producing sectors accelerated from the first to the second half of the 1990s.

All in all, these previous studies suggest that ICT has had some positive effect on economic growth in the euro area, that this effect tended to increase over the 1990s, but remained relatively limited. The literature reviewed, however, gives no clear verdict as regards the existence of positive spillover effects.

3 Sectoral developments

To assess the importance of ICT for the production side of the economy in the euro area, this section examines developments in output and productivity growth at the sectoral level, in the ICT sector in particular. The ICT sector as a whole is defined as consisting of ICT producing and ICT using sectors (see Table 1). The classification of ICT producing industries closely follows that of the OECD (2000a), while the classification of ICT using sectors, because any positive spillover effects from the use of ICT should become apparent in sectors other than those producing ICT. ICT using sectors are defined as those which have a relatively high ratio of ICT investment to industry output and a relatively high share in the overall ICT capital stock. This is admittedly somewhat arbitrary. Indeed, 'non-ICT using sectors will also use ICT to some extent, and it could be argued that even a limited use of ICT could cause a clear improvement in the efficiency of the production process. The ICT using sectors distinguished here might thus be seen as only a rough measure for assessing the importance of ICT use in the economy as a whole.

Table 1: Classification of ICT producing and using industries

ICT producing sector, manufacturing

Office, accounting and computing machinery (code 30) and radio, television and communication equipment (code 32).

ICT producing sector, services

Post and telecommunications (code 64) and computer and related activities (code 72).

ICT using sector, manufacturing:

Chemicals and chemical products (code 24), electrical machinery and apparatus, not elsewhere classified (code 31), and medical, precision and optical instruments (code 33).

ICT using sector, services:

Financial intermediation (code 65), insurance and pension funding (code 66), activities related to financial intermediation (code 67), renting of machinery and equipment (code 71), research and development (code 73), and 'other business activities' (code 74).

Note: codes in brackets are from the international standard industry classification, revision 3. Only about half of the category 'other business services' qualifies as ICT using. Therefore a 50% split was applied for this category.

A shift share analysis has been carried out in order to determine the contribution of a given sector to overall productivity growth more precisely. This method implies that ALP for the total economy (P) can be written as the sum of the ALP contributions of individual sectors (i) weighted with their labour share $(L_i/L=S_i)$:

$$P = Y/L = \sum_{i} (Y_{i}/L_{i}) (L_{i}/L) = \sum_{i} (P_{i} * S_{i})$$
(1)

In a time perspective this equation can be rewritten as:

$$\Delta P_{(t)} = \left(\sum_{i} (\Delta P_{i(t)} * S_{i(t-1)}) + \sum_{i} (P_{i(t-1)} * \Delta S_{i(t)}) + \sum_{i} (\Delta P_{i(t)} * \Delta S_{i(t)}) \right)$$
(2)

where Δ takes differences across time. The first term on the right hand side of the equation is the so-called 'within' effect. It measures the contribution of the ALP growth *within* the individual sector to overall productivity growth. It can be interpreted as the counterfactual rate of productivity growth in the absence of changes in the production structure. The second and third terms represent the contribution of productivity growth from changes in the employment shares between sectors. A shift of employment from sectors with low productivity levels to sectors with high productivity levels will show a positive 'static' effect (the second term). The third term measures the contribution of a shift in employment shares of a sector multiplied by its productivity growth. This so-called 'dynamic' effect will be positive if the share of a sector that shows above average productivity growth increases. The total contribution of any sector to overall ALP growth is obtained by summing the separate components for the sector.

Data on gross value added and employment (in persons) at a detailed sectoral level are available for five euro area countries: Germany, France, Italy, the Netherlands, and Finland (comprising about 81% of euro area gross value added). The data from these five countries are used to construct an estimate for gross value added and employment by sector for the euro area (see Appendix 1 on data sources and aggregation methods for more information).⁵ Unfortunately, data for all five countries are available only for the period 1991 to 1998. Due consideration should be given to the fact that the results may be influenced by the particular cyclical position of countries in these years.

⁴ The classification used here is somewhat less detailed in that both the OECD and Van Ark distinguish sectors at the three-digit sectoral level.

 $^{^{5}}$ The euro area economic structure may differ from the structure implicit in the euro area estimate presented here, which is based on only a subset of countries. Arguably, the availability of statistics correlates positively with the degree of countries' economic development, which in turn could apriori be assumed to positively correlate with the degree of ICT penetration in the economy. This would imply that there might be an upward bias in the estimates of the contribution of ICT to ALP growth. However, here, and especially in the growth accounting exercise, important euro area producers of ICT (Ireland and Finland) are not taken into account – due to lack of data. Furthermore, the aggregation of gross value added is not fully harmonised across euro area countries, as use is made of both chain-weighted and fixed-weight aggregates. Moreover, considerably different price indicators are used, including hedonic deflators. Here (and in the following subsection) these factors are not taken into account. All this implies that there is probably a bias in the euro area estimate as presented in this study, the precise size and direction of which are however unknown.

Table 2 presents for the euro area, for each ICT sector, the output share and output growth, the employment share and employment growth, as well as ALP growth. The manufacturing and business services sectors as well as the total economy have been added as benchmarks for the developments in the ICT sectors. Appendix 2 gives an overview of the individual country results. Table 3 presents the contributions to overall ALP growth of the ICT sectors.

Table 2 Sectoral developments in the euro area¹

	Share i	n	Growth	in	Share	in	Growth	in	Growth	in	
	nominal value	added	real value ad	real value added		employment		employment		ALP	
	1991	1998	1991-1998	1995-1998	1991	1998	1991-1998	1995-1998	1991-1998	1995-1998	
ICT producing sectors, manufacturing	1.0%	0.8%	6.6%	12.3%	0.9%	0.6%	-5.2%	-1.8%	12.5%	14.3%	
ICT producing sectors, services	3.3%	3.8%	5.9%	8.9%	2.4%	2.4%	-0.2%	0.9%	6.0%	7.9%	
ICT using sectors, manufacturing	4.2%	3.6%	0.9%	1.3%	3.6%	2.9%	-3.0%	-1.2%	4.0%	2.4%	
ICT using sectors, services	10.5%	11.2%	2.4%	3.3%	6.8%	8.1%	2.2%	3.2%	0.1%	0.1%	
Manufacturing	23.3%	20.8%	0.8%	1.5%	23.9%	20.2%	-2.4%	-0.6%	3.4%	2.1%	
Business services	44.2%	47.7%	2.3%	3.0%	34.9%	38.3%	1.2%	2.1%	1.1%	0.9%	
Total economy	100.0%	100.0%	1.6%	2.0%	100.0%	100.0%	-0.1%	0.6%	1.7%	1.3%	

¹ estimate based on Germany, France, Italy, the Netherlands, and Finland comprising about 81% of euro area gross value added

Source: own calculations using data from STAN OECD database, Groningen University ICT database, and Statistics Netherlands Note: due to the rapid decline of measured prices in the ICT producing manufacturing sector its share in nominal value added declined, despite high rates of growth in real value added.

Manufacturing and business services include the ICT sectors.

	je laboul	producti	vity grow			1		
(as percent of total ALP growth)								
	Overall con	Overall contribution		ct	Static shift	effect	Dynamic shift effect	
	1991-1998	1995-1998	1991-1998	1995-1998	1991-1998	1995-1998	1991-1998	1995-1998
ICT producing sectors, manufacturing	3.4%	7.5%	7.5%	9.5%	-1.8%	-1.4%	-2.3%	-0.7%
ICT producing sectors, services	12.6%	23.0%	12.7%	22.0%	-0.1%	0.8%	0.0%	0.2%
ICT using sectors, manufacturing	2.3%	1.8%	10.0%	7.2%	-5.8%	-5.0%	-1.8%	-0.4%
ICT using sectors, services	16.0%	22.0%	0.8%	1.2%	15.1%	20.7%	0.1%	0.1%

Table 3 Contribution to average labour productivity growth in the euro area¹

¹ estimate based on Germany, France, Italy, the Netherlands, and Finland comprising about 81% of euro area gross value added Source: own calculations using data from STAN OECD database, Groningen University ICT database, and Statistics Netherlands

Table 2 shows that the *ICT producing sectors*, both in manufacturing and services, were highly dynamic in terms of growth rates of real value added and ALP. Moreover, there was a clear pick-up in the growth rates of both variables in the second half of the 1990s. However, the size of the ICT producing sectors is small, with a share of less than 5% in total nominal value added and roughly 3% in total employment, implying that their impact on activity developments in the euro area as a whole is limited. Nevertheless, the contribution to overall ALP growth is noticeable, being about one sixth (0.3 percentage point) of total euro area ALP growth over the period 1991-1998, and even one third (0.4 percentage point) in the years 1995-1998 (see Table 3).⁶ The own dynamics of the sector

 $[\]overline{}^{6}$ Strictly speaking a shift share analysis can not be applied to chain-weighted data, as chain-weighted data are not additive over the sectors (see e.g. Whelan 2000). In our sample the data for France are chain-weighted, however, as explained in footnote 3, in the euro area aggregate this has not been taken into account, implying that the euro area estimates used here are additive over the sectors.

(the 'within' effect) have been paramount in explaining this contribution. Shift effects played only a minor role and even reduced the total contribution of the ICT producing manufacturing sector.

As to the *ICT using sectors*, there has been no clear increase in the growth rates of real value added or ALP. In the ICT using services sector measured ALP growth has even been close to zero during the 1990s. The ICT using services sector has also been the one with the highest employment growth, reaching somewhat over 3% in the second half of the 1990s. Indeed, ICT producing sectors and the ICT using manufacturing sector had hardly any or even negative employment growth over the same period. The absence of stronger dynamics in the ICT using sectors than on average in the manufacturing and the business services sectors suggests that over the period examined positive spillover effects from the use of ICT have been limited if present at all. As to its contribution to overall ALP growth, the within effect in the ICT using manufacturing sector has been substantial. However, due to the decline in employment share, the overall contribution of this sector has been small. By contrast, the ICT using services sector has mainly contributed to overall productivity growth on account of a substantial increase in the employment share of this sector over time.

A direct comparison of the results for the euro area with those for the United States (shown in Table 4) reveals the following.⁷ First, the growth rates of ALP of the ICT producing manufacturing sectors in the euro area seem roughly comparable to those in the United States. This should be seen against the background of statistical problems, which adds to the measured difference between the euro area and the United States on account of more rapidly declining deflators for ICT producing manufacturing sectors in the United States. The hedonic method used in the United States to separate price and quality changes tends to lead to lower measured price increases and higher measured output growth than the approaches used in most euro area countries, where only France uses the hedonic approach (see also appendix 4 'Measurement problems related to productivity'). In the period 1995-1998, for example, the decline in the value added deflator for the ICT producing manufacturing sector was 12.5% per year on average in the United States and 7.2% in the euro area. The difference of more than 5 percentage points, being a rough measure of the possible impact of measurement errors clouding the comparison, almost completely accounts for the differences in measured ALP growth. Bearing in mind this

⁷ The construction of ICT producing services sector data for the United States was hampered by classification problems. A direct comparison with the euro area data therefore seemed not justified. Moreover, no shift-share analysis for the United States is presented here, as the use of chain-weighted indices makes such an analysis impossible without the introduction of fairly restrictive assumptions. See also footnote 6.

caveat is important in interpreting Tables 3 and 4, which show ALP as measured by the currently available statistics, in an attempt to let these data 'speak for themselves'.

Second, in the United States the high ALP growth was accompanied by an above average increase in employment in the ICT producing manufacturing sector. This contrasts sharply with developments in the euro area over this period. This could be indicative of structural impediments to growth in the euro area, such as barriers to the creation of firms, for example resulting from the regulatory framework or the relative dearth of venture capital, to inflexible labour markets or to a lack of human capital in the ICT producing manufacturing sector.

Third, the share of the ICT producing manufacturing sector in total nominal value added in the US is at 1.8% in 1998 more than twice as high as the corresponding share in the euro area. This implies that the impact of this dynamic sector on economy-wide developments is clearly stronger in the United States than in the euro area. Moreover, the output share of all ICT (producing and using) sectors taken together clearly increased in the United States from 19.6% in the first half of the 1990s to 22.8% in the second half, whereas it was more or less stable in the euro area at 19.0% and 19.4% respectively.

Fourth, in the United States, as in the euro area, the dynamics of the ICT using sectors in terms of value added and ALP appear not to be particularly strong when compared with the benchmark sectors (total manufacturing and total business services respectively). However, in the United States there has been an increase in ALP growth in the ICT using sectors from the first to the second half of the 1990s, which was larger than the increase in the benchmark sectors. This is consistent with the finding of Stiroh (2001a, 2001b) that the industries in the United States which recorded an acceleration in ALP in the second half of the 1990s were more intensive users of ICT capital.

	Share	Share in Growth in				Share in		Growth in		Growth in	
	nominal value	added	real value ad	real value added		employment		employment		ALP	
	1991	1998	1991-1998	1995-1998	1991	1998	1991-1998	1995-1998	1991-1998	1995-1998	
ICT producing sectors, manufacturing	1.5%	1.8%	20.9%	25.6%	1.0%	0.9%	1.4%	3.5%	19.2%	21.3%	
ICT producing sectors, services	4.8%	5.2%	4.9%	5.4%	3.0%	3.3%	3.0%	4.1%	1.9%	1.2%	
ICT using sectors, manufacturing	3.4%	3.1%	3.0%	4.2%	2.2%	1.8%	-0.9%	0.2%	3.9%	4.0%	
ICT using sectors, services	9.9%	12.7%	4.6%	7.4%	8.4%	9.3%	3.3%	4.4%	1.3%	2.8%	
Manufacturing ¹	17.4%	16.3%	4.5%	4.0%	14.9%	13.4%	0.3%	0.6%	4.1%	3.4%	
Business services ¹	49.1%	53.2%	4.7%	6.4%	43.8%	45.9%	2.5%	2.9%	2.1%	3.4%	
Total economy	100.0%	100.0%	3.5%	4.1%	100.0%	100.0%	1.8%	2.1%	1.7%	2.0%	

Table 4 Sectoral developments in the US

¹ Manufacturing and business services include the ICT sectors.

Source: own calculations using data from STAN OECD database, Groningen University ICT database, and Statistics Netherlands

Overall, one may conclude that output and ALP growth in ICT *producing* sectors in the euro area were clearly higher than in other sectors of the economy. This points to a positive impact of ICT on economic growth. However, the size of these sectors is relatively small implying that there has so far been only a limited impact on overall economic developments. Nevertheless, the aggregate contribution of ICT sectors to total ALP growth was noticeable in the euro area in the period 1991-1998. The fact that the growth rates of ALP in the ICT *using* sectors did not rise appreciably faster than in the non-ICT using sectors casts doubt, for the time being, on the existence of positive spillover effects from the use of ICT in the euro area. Moreover, the relatively low growth rates of employment in the ICT producing sectors in the euro area could be indicative of a lack of flexibility in the product, labour, and financial markets.

4 Contribution of new technologies to economic growth

To assess the contribution of ICT capital to economic growth and to estimate the development of TFP, a standard growth accounting exercise has been carried out. The growth accounting framework was pioneered by Solow (1957) and further developed by Jorgenson and associates (e.g. Jorgenson and Griliches, 1967; Jorgenson et al 1987). The framework used here is similar to that used in Oliner and Sichel (2000). In a growth accounting framework, the growth rate of output (\dot{Y}) is equal to the weighted growth rates of labour input (\dot{L}) and capital input (\ddot{K}) , plus growth in total factor productivity (TFP). The following formula has been used here:

$$\dot{Y} = \alpha_{\rm L} \dot{L} + \sum_{i} \alpha_{\rm Ki} \dot{K}_{i} + TFP$$
(3)

Time subscripts have been suppressed for simplicity of notation. Labour input growth is measured in total hours worked (Appendix 1 provides a more detailed overview of the data used). The share of labour (α_L) can be calculated from the wage share in gross value added (which can be directly extracted from the national accounts) adjusted for the imputed wage income of the self-employed and varies over time. Due to data limitations, no adjustment has been made for the quality of labour in this exercise. As to capital inputs, a distinction is made between the contribution of ICT capital and of other, non-ICT capital to output. In all, six categories of capital have been distinguished. ICT capital consists of the stock of information equipment (including computers), the stock of

software, and the stock of communications equipment. Non-ICT capital consists of the stocks of 'other machinery and equipment', transport equipment and non-residential construction. Capital stock estimates have been constructed using the perpetual inventory method, which uses the past pattern of real investment together with assumptions on service lives and age-efficiency patterns of the different types of capital goods (see also Appendix 1). The sum of the shares of the various types of capital is assumed to be equal to 1- α_L , a standard assumption in this kind of exercise reflecting constant returns to scale. The shares of the different types of assets in total capital input are based on the user cost of capital, i.e. the gross rate of return that must cover the internal rate of return (assumed common to all capital)^s, the depreciation rate, and the capital gain/loss of the specific capital good. Tax considerations were not taken into account, but the impact of taxes on the user cost of capital is assumed to be captured by the internal rate of return.

It is important to mention the following caveats. First, the growth accounting exercise used here relies on the Cobb-Douglas framework, with all production factors entering as mutually complementary. ICT is thus treated as just another capital good, one that is not different from others in terms of its impact on production. However, some observers have argued that ICT will have more fundamental implications for the organisation of work, making ICT a substitute of rather than a complement to other types of capital. Extending the framework to allow for ICT being a substitute is however beyond the scope of this paper.

Second, TFP growth as estimated here reflects a Hicks-neutral shift of a production possibility function over time. The estimates of TFP growth would be biased to the extent that technical progress is not neutral.

Third, the implicit assumption made here is that ICT affects economic growth immediately. However, some have argued that the benefits of ICT for economic growth will only be observed with a lag (e.g. David 1990).

Fourth, the growth accounting framework departs from the assumption of maintained equilibrium. In periods of structural changes, this assumption does not hold. Arguably, the increased use of ICT could be seen as such a structural change. Kiley (2000), for instance, has tried for the United States to incorporate costs of adjustment and concluded that the inclusion of adjustment costs can have large effects on the growth-accounting exercise when a new investment good is introduced - such as ICT. The contribution of ICT to economic growth could consequently be constrained for a prolonged period by the large adjustment costs required to incorporate a new investment good into the economy's

⁸ This assumption is not consistent with the views of some, who argue that ICT accounts for exceptionally high returns to investment.

capital stock. Eventually, however, the impact of ICT boosts long-run growth in his model as well.

Fifth, ideally TFP growth as derived here (i.e. as a residual term) should reflect the increase in efficiency in the economic process. Hence, any positive spillover effects from ICT investment should result in an increase in the estimate of TFP growth.^o However, as TFP growth is a residual term it captures all elements not included in the growth rates of capital and labour input, and thus also reflects the impact of omitted variables such as the quality of labour and any biases due for example to measurement problems." It is therefore difficult to draw any firm conclusions from changes in measured TFP growth for the development of overall efficiency.

Using equation (3), the contribution of ICT capital to output and ALP growth has been determined and estimates have been made of TFP growth for the period 1991-1999. Usually TFP growth shows a pro-cyclical pattern. However, in view of the difficulties to separate trend from cycle, especially over short time periods, no attempt has been made to distinguish trend productivity growth from cyclical effects. Rather, the focus is on actual developments in the course of the 1990s." For the euro area, there is a scarcity of national accounts data on ICT investment. However, the euro area estimates presented below are based on national accounts data from four countries (Germany, France, Italy and the Netherlands) which together comprise almost 80% of euro area gross value added.¹²

Table 5 shows the decomposition of growth of total real value added for the euro area estimate (using the implicit deflators from national data). Appendix 3 provides an overview of the individual country results. Table 5 presents absolute contributions to output growth as well as relative contributions, which represent the contribution relative to total growth. It appears that the relative contribution of ICT capital to growth has increased from 13% in the first half of the 1990s to 21% in the second half, largely due to software and, to a lesser extent, to information equipment. By contrast, the contribution of communications equipment investment has been remarkably stable over time. The increased contribution of ICT capital to the growth of real value added has been accompanied by a decline not only in the relative but also in the absolute contribution of

⁹ Some researchers suggested on the basis of US data that technological change that is embodied in new ICT capital goods, which is not adequately reflected in the official price indices, would bias downward the measured growth of effective ICT capital stock (e.g. Sakellaris and Wilson 2001). In the current analysis, the effect of such embodied technological change is not identified and should show up in the overall estimate of TFP growth.

¹⁰ As Triplett (2001) pointed out "if output and computer inputs are correctly measured, the new things that computers do will not show up in economic statistics in the form of an enhanced growth of [T]FP", ¹¹ With the notable exception of Gordon (2000), this has been the approach taken in most of the US literature as well. For instance, Stiroh (2001b) noted: "It is important to point out that there is no attempt to cyclically adjust the data; all analysis is done using actual data as reported by BEA. [...], it is quite difficult to separate trend and cyclical components, particularly when the data end in the middle of the cycle, as is currently the case." ¹² See footnote 3.

non-ICT capital. The contribution of total hours worked to output growth has turned positive in the second half of the 1990s to 22% in relative terms, following a substantial decrease in hours worked in the first half of the decade. The estimates give no indication of an increase in TFP growth in the course of the 1990s. On the contrary, TFP growth declined markedly in both absolute and relative terms from the first to the second half of the decade. In this context, it should be noted that the measure of TFP growth used here implicitly also includes the impact of changes in the quality of labour. In a situation of relatively low-skilled and inexperienced workers, the quality of labour input may grow at a slower pace than in a situation in which these people would not have entered employment. Hence, the decrease in measured TFP growth in the euro area in the second half of the 1990s is not necessarily a negative sign as it probably also partly reflects the absorption of previously unused supply of labour.

	Absolute contri (percentag	bution to growth ge points)	Relative contri (as a percer	bution to growth nt of total)
	1991-1995	1996-1999	1991-1995	1996-1999
ICT capital	0.20	0.40	13	21
 Information equipment 	0.09	0.15	6	8
- Software	0.05	0.18	4	9
- Communications equipment	0.06	0.07	4	3
Other capital	0.57	0.47	37	24
Total hours worked	-0.66	0.43	-43	22
ТГР	1.41	0.63	92	33
	annual average pe	ercentage growth		
Gross real value added	1.5	1.9	100	100

Table 5 Decomposition of euro area output growth¹

¹ estimate based on Germany, France, Italy, and the Netherlands, comprising about 79% of euro area gross value added Source: own calculations based on data from OECD and national accounts

A closely related exercise focuses on the decomposition of ALP growth, whereby the growth in total hours worked is subtracted from the growth in output and from the growth in the various inputs. In this decomposition, ALP growth reflects increases in the amount of capital available per hour worked (capital deepening) and in the growth rate of TFP. Table 6 presents the results. According to these estimates, ALP growth decreased from 2.4% in the first to 1.3% in the second half of the 1990s. This decrease can be attributed to both a decline in TFP growth, and to a decrease in the rate of capital deepening of non-ICT capital. By contrast, ICT capital deepening accelerated over the same period from 10% to 28% in terms of relative contributions.

	Absolute contri (percentag	bution to growth ge points)	Relative contri (as a percer	bution to growth it of total)
	1991-1995	1996-1999	1991-1995	1996-1999
ICT capital deepening	0.25	0.36	10	28
 Information equipment 	0.11	0.14	4	11
- Software	0.07	0.16	3	13
- Communications equipment	0.07	0.06	3	4
Other capital deepening	0.77	0.32	32	24
ТҒР	1.41	0.63	58	48
	annual average pe	ercentage growth		
ALP	2.4	1.3	100	100

Table 6 Decomposition of euro area average labour productivity growth¹

¹ estimate based on Germany, France, Italy, and the Netherlands, comprising about 79% of euro area gross value added Source: own calculations based on data from OECD and national accounts

Table 7 compares the results of this study with those of similar studies for the United States, attempting to take into account the methodological differences with regard to price deflators.¹⁷ In particular, for the euro area estimate the US deflator for information equipment is substituted for the national one. Comparisons are further hampered by differences in the output concept used: while this study focuses on GDP, the studies on the United States cited refer to private sector output. Nevertheless, some interesting results appear.

It appears that the differences in the contribution of ICT capital to growth between the euro area and the United States are not very different, contrary to conventional wisdom. Table 7 suggests that the most significant difference between the euro area and the United States is in the development of TFP growth, which decreased in the euro area, but increased in the United States. To some extent the deceleration in TFP in the euro area may be explained by a deceleration in labour quality, as explained above. However, according to the studies cited in Table 7, in the United States the contribution of labour quality to output growth also declined slightly (about 0.1 percentage point) from the first to the second half of the 1990s. The results from the comparison are consistent with the finding of Stiroh (2001b) that increased ICT-use did not cause TFP to accelerate in the case of the United States (even though a positive impact of ICT-use on ALP is discernible at the sectoral level). However, he did find that TFP growth increased above average in the United States (even though a positive impact of ICT-use of this sector in the United States (even though a positive impact of ICT-use of the sector in the United States (even though a positive impact of ICT-use of the sector average in the ICT producing sector. This suggests that the larger size of this sector in the United

¹³ Because of the use of US-based alternative deflators, the growth contributions for the euro area in Table 7 differ from the results in Table 5. See appendix 4 for a discussion of the many methodological and statistical difficulties surrounding international productivity comparisons.

States is one of the explanations for the difference in TFP growth between the euro area and the United States.

Admittedly, the adjustment made here to the euro area estimate is rather crude. Whilst it is true that different approaches to quality adjustment may significantly impact on information equipment deflators, a number of other factors also need to be taken into account. For instance, the price indices in each country may reflect a different mix of investment goods, which suggests that using a US-based alternative deflator is far from ideal. Moreover, especially when investment goods are imported, a currency conversion could be warranted. Note also that adjustment of the deflators for information equipment could have an impact on GDP and ALP. However, the analysis in appendix 4 suggests that any such impact on the growth rate of aggregate output is only very limited (even though the effect on GDP expenditure components and measured real value added in individual sectors likely is much more substantial).

absolute contributio	n to growth (perc	entage point)									
	Country	Period	Contributions to output growth ¹								
			IT equipment ²	Software	Comm. equipment	TFP growt					
This study	euro area	1991-1995	0.26	0.05	0.06	1.24					
		1996-1999	0.49	0.18	0.07	0.29					
Oliner/Sichel	United States	1991-1995	0.25	0.25	0.07	0.92					
		1996-1999	0.63	0.32	0.15	1.47					
Jorgenson/Stiroh	United States	1990-1995	0.19	0.15	0.06	0.73					
-		1995-1998	0.46	0.19	0.10	1.24					

Table 7 Comparison of this study with studies on the United States absolute contribution to growth (percentage point)

¹This study: GDP; Oliner/Sichel: nonfarm business sector output; Jorgenson/Stiroh: private domestic output

² Using an alternative US based deflator for IT equipment to increase comparability

³ Including changes in labour quality

In summary, the results of the standard aggregate growth accounting exercise suggest that the importance of ICT capital accumulation for economic growth in the euro area has increased in the second half of the 1990s. The euro area is thus experiencing positive growth effects of ICT. The size of the contribution even appears to be not very different to that in the United States if one attempts to allow for the difference in price deflators for IT equipment. However, the data available do not point in the direction of significant positive spillover effects of ICT investment on the rest of the economy in the euro area, since, according to the estimates presented here, TFP growth has been declining rather than increasing in the course of the last decade.

5 Concluding remarks

The analysis of output and productivity developments in the euro area undertaken in this paper suggests that in the period up to 2000 there were only very limited, if any, positive spillover effects of the use of ICT on overall efficiency in the economic process. Hence, there is no reason to believe that the growth rate of potential output of the euro area would have risen significantly in recent years as a result of ICT developments. However, the data do show clear evidence of an increased contribution of ICT to overall economic growth. It has often been argued that the effects of ICT on TFP may occur with a certain delay. Looking forward, it thus might be that the forces of technological change are already discretely operating in the background. This would also imply that the uncertainties surrounding estimates of medium-term developments in potential output growth might have become skewed to the upside.

Many observers have attributed the differences in measured productivity growth (whether it is ALP or TFP growth) between the euro area and the United States in the second half of the 1990s, to a large degree to differences in the contribution of ICT capital. However, the contribution of ICT capital to output growth has not been much different in the euro area than in the United States, if one attempts to allow for the effects of different deflation techniques. This suggests that other factors are likely to account for the largest part of the observed, substantial, difference in TFP growth, including differences in production structure and possibly also flexibility of product, labour, and financial markets.

References:

Aizcorbe, A., C. Corrado and M. Doms (2000), 'Constructing Price and Quantity Indexes for High Technology Goods', Federal Reserve Board of Governors, mimeo.

Bailey, M.N. and R.Z. Lawrence (2001), 'Do We Have a New E-conomy?', NBER, working paper No. 8243.

Bureau of Economic Analysis (1999), *Fixed Reproducible Tangible Wealth in the United States, 1925-94*, Washington DC.

Cecchetti, S.G. (2000), 'Early Warning Signs of the US Productivity Pickup: Implications for Europe', Ohio State University, mimeo.

Colecchia, A. and P. Schreyer (2001), 'ICT Investment and Economic Growth in the 1990s: is the United States a Unique Case? A Comparative Study of nine OECD Countries', OECD, mimeo;

CPB (2000), *ICT en de Nederlandse economie. Een historisch en internationaal perspectief*, Werkdocument No. 125, Den Haag.

Daveri, F. (2001), 'Information Technology and Growth in Europe', University of Parma/IGIER working paper.

David, P.A. (1990), 'The Dynamo and the Computer: A Historical Perspective on the Modern Productivity Paradox', *American Economic Review*, 80(2), pp. 355-361.

Deutsche Bundesbank (2001), 'Appendix: Problems of International Comparisons of Growth – A Supplementary Analysis', *Monthly Report, May*, pp. 39-43.

European Central Bank (2001), 'New Technologies and Productivity in the Euro Area', *ECB Monthly Bulletin, July*, p.p. 37-48.

Fagan, G., J. Henry and R. Mestre (2001), 'An Area-Wide Model (AWM) for the Euro Area', European Central Bank, working paper No. 42.

Fraumeni, B.M.(1997), 'The Measurement of Depreciation in the US National Income and Product Accounts', *Survey of Current Business*, 77, p.p. 7-23.

Gordon, R.J. (2000), 'Does the "New Economy" Measure up to the Great Inventions of the Past?', *Journal of Economic Perspectives*, 14(4), pp. 49-77.

Jalava, J. and M. Pohjola (2001), 'Economic Growth in the New Economy – Evidence from Advanced Economies', UNU/WIDER, working paper No. 2001/5.

Jorgenson, D.W. and Z. Griliches (1967), 'The Explanation of Productivity Change', *Review of Economic Studies*, 34, pp.249-280.

Jorgenson, D.W., F.M. Gollop and B.M. Fraumeni (1987), *Productivity and US Economic Growth*, Cambridge MA., Harvard University Press.

Jorgenson, D.W. and K.J. Stiroh (2000), 'Raising the Speed Limit: U.S. Economic Growth in the Information Age', *Brookings Papers on Economic Activity*, 1, pp. 125-211.

Jorgenson, D.W. (2001), 'Information Technology and the US Economy', *American Economic Review*, 91(1), pp. 1-32.

Kiley, M.T. (1999) 'Computers and Growth with Costs of Adjustment: Will the Future look like the Past?' Federal Reserve Board of Governors', Finance and Economics Discussion Series 1999-36.

Lequiller, F. (2001), 'The New Economy and the Measurement of GDP Growth', INSEE, working paper G 2001/01.

McKinsey Gobal Institute (2001), 'US Productivity Growth 1995-2000: Understanding the Contribution of Information Technology relative to other Factors', Washington DC.

Mairesse, J., G. Cette et Y. Kocoglu (2000), 'Les technologies de l'information et de la communciation en France: diffusion et contribution à la croissance', *Economie et Statistique*, 339-340, p.p. 117-146.

Nordhaus, W. (2001), 'Productivity Growth and the New Economy', NBER, working paper no. 8096.

OECD (1999), Purchasing Power Parities and Real Expenditures: 1996 Results, Paris.

OECD (2000a), OECD Information Technology Outlook 2000, Paris.

OECD (2000b), Labour Force Statistics 2000, Paris.

Oliner, S.D. and D.E. Sichel (2000), 'The Resurgence of Growth in the Late 1990s: Is Information Technology the Story?', *Journal of Economic Perspectives*, 14(4), pp. 3-22.

Oulton, N. (2001), 'ICT and Productivity Growth in the UK', Bank of England Working Paper No. 140, July.

Roeger, W. (2001), 'The Contribution to Information and Communication Technologies to Growth in Europe and the US: A Macroeconomic Analysis', European Commission DG-ECFIN, Economic Papers No. 147.

Sakellaris, P. and D. Wilson (2001), 'Embodied Technical Change: Plant-level Estimates', University of Maryland, mimeo.

Scarpetta, S., A. Bassanini, D. Pilat and P. Schreyer (2000), 'Economic Growth in the OECD Area: Recent Trends at the Aggregate and Sectoral Level', OECD, Economics Department Working Papers No. 248.

Schreyer, P. (2000a), 'The Contribution of Information and Communication Technology to Output Growth: A Study of the G7 Countries', OECD, STI working paper 2000/2.

Schreyer, P. (2000b), 'Information and Communication Technology and the Measurement of Volume Output and Final Demand – A Five Country Study', *Economics of Innovation and New Technology*.

Solow, R. (1957), 'Technical Change and the Aggregate Production Function', *Review of Economics and Statistics*, 39(3), pp. 65-94.

Stiroh, K.J. (2001a) 'Information Technology and the U.S. Productivity Revival: What do the Industry Data say?' Federal Reserve Bank of New York Staff Reports, No. 115.

Stiroh, K.J. (2001b), 'Are ICT Spillovers Driving the New Economy?', Federal Reserve Bank of New York, mimeo.

System of National Accounts (1993).

Triplett, J. E. (2001), 'Did the US have a New Economy?', paper prepared for 9th conference on National Accounting of the Association de Comptabilite Nationale, Paris, November 2001, mimeo.

Van Ark, B. (1996), 'Issues in Measurement and International Comparison Issues of Productivity - An Overview' *OECD proceedings, Industry Productivity: International Comparison and Measurement Issues*, Paris, pp. 19-48.

Van Ark, B. (2000), 'De vernieuwing van de oude economie. Nederland in een internationaal vergelijkend perspectief', in Soete, L. (ed.) 'ICT en de niewe economie', *Preadviezen van de Koninklijke Vereniging voor Staathuishoudkunde*, Utrecht, pp.21-61.

Van Ark, B. (2001), 'The Renewal of the Old Economy: An International Comparative Perspective', OECD, STI working paper 2001/5.

Van der Wiel, H., (2001), 'Does ICT boost Dutch productivity growth?', CPB Document No. 016.

Wellink, A.H.E.M. and R.M. Albers (2001), 'De nieuwe economie in het perspectief van de 20^e eeuw' in: S. Brakman and A. van Witteloostuijn eds., *Jonge wijn in oude zakken: op weg naar een nieuwe economie*, Amsterdam, Arbeiderspers.

Whelan, K. (2000), 'A Guide to the Use of Chain Aggregated NIPA Data', 'Federal Reserve Board of Governors', Finance and Economics Discussion Series 2000-35.

Wyckoff, A.W. (1995), 'The Impact of Computer Prices on International Comparisons of Labour Productivity', *Economics of Innovation and New Technology*, 3, pp. 277-293.

Appendix 1 Data sources and aggregation methods

Data for sectoral developments

The main data source used is the OECD STAN database, which contains data on a detailed (two-digit ISIC rev.3) sectoral level for gross value added and employment. This is still under development and database as yet not published (see http://www.oecd.org/dsti/stat-ana/stats/). In some cases, data from this database are not available or not sufficiently detailed. In those cases, the ICT database of Groningen University (see http://www.eco.rug.nl.GGDC/) has been used as an additional source of information. Moreover, for the Netherlands, use has been made of detailed employment accounts and supply and use tables of Statistics Netherlands. Employment includes selfemployed persons in all cases.

Data for growth accounting

<u>Labour hours</u>: total employment data (in persons) were taken from the OECD (2000b) and average working hours from Scarpetta et al (2000). The series on average working hours was extended by one year (to 1999) on the assumption that the trend of the three preceding years was sustained.

<u>Labour share</u>: The share of labour (α_L) is calculated from the national accounts, by adding to the share of the employees (which can be directly read from the national accounts) the share of the self-employed, assuming that the share of the latter is proportionally equal to the share of the employees.

Capital stocks: To construct the capital stocks the following formula has been used:

$$\mathbf{K}_{\mathrm{it}} = \sum_{b=0}^{b=t} \Phi_{\mathrm{itb}} \mathbf{I}_{\mathrm{ib}}$$

Where I_{ib} represents real investment at time b of capital good i, and Φ_{itb} the economic efficiency at time t of investments at time b of capital good i. Φ_{itb} in turn is calculated from the formula:

$$\Phi_{itb} = (m_i - a_i) / (m_i - \beta_i a_i)$$

with m_i the average service lifes set equal to those of the US Bureau of Economic Analysis (1999), a_i being the age of the vintage, and β_i the decay parameters set at 0.95 for ICT capital and 0.8 for other types of capital, following CPB (2000) on the assumption that the decay in efficiency of ICT is relatively very limited until the asset is retired at the end of the service life. However, assuming a common decay parameter to all types of capital (including ICT capital) does not significantly alter the results. Note that the service life for software has been set at 4 yours, based on assumed service lifes for pre-packaged and own account software in Oliner and Sichel (2000).

The investment data necessary to build capital stocks are based on national accounts (ESA95) of Germany, France, Italy, and the Netherlands. In the case of Germany, the ICT investment series had to be backcast by assuming that the growth pattern of the other three countries was representative for Germany as well. In addition, the structural break due to German re-unification has been corrected by applying West-German growth rates to all German levels back in time. Moreover, in the case of non-ICT investment, some series had to be backcast by applying growth rates of ESA79 data to the ESA95 time series in order to construct long-enough time series. The investment data were aggregated to yield estimates of euro area investment in the different types of capital goods distinguished.

<u>Share of capital</u>: The income share for each type of capital is calculated from the following equation:

 $\alpha_{Kit} = (c_{it} K_{it}) / (p_{yt} Y_t)$

where Y is real gross value added, and c_{it} the user costs of capital, which are calculated by using the following formula:

 $c_{it} = (r_t + \delta_{it} - \pi_{it}) P_{kit}$

where δ_{it} represents the depreciation rate, which is taken from the tables of the US Bureau of Economic Analysis (1999) as in Oliner and Sichel (2000), π_{it} is the expected capital gain/loss and is calculated as a three-year moving average of the annual price change of the capital good (following CPB 2000 and Oliner and Sichel 2000), P_{kit} is the price of the respective capital good, r represents the nominal rate of return and is assumed to be equal over all types of stocks of capital goods. The depreciation rate for software has been derived in a consistent manner with BEA's depreciation estimates for the other types of capital goods. The rate of return is thus calculated for each year as the expost return from the equation:

$$\sum_{i} \left(\left(r_{t} + \delta_{it} - \pi_{it} \right) P_{kit} K_{it} \right) / \left(p_{yt} Y_{t} \right) = 1 - \alpha_{Lt}$$

Aggregation methods

Where appropriate, purchasing power parities were used to compute euro area aggregates, in accordance with standard practices for cross-country comparisons of economic growth (Van Ark 1996). 1996 purchasing power weights as calculated by the OECD (1999) were applied here. In particular, the expenditure PPPs were matched to the particular sector and investment category distinguished (the so-called 'proxy' PPP approach). This choice is motivated by the need for a conversion factor which takes cross-country differences in price levels and relative price differences among expenditure categories into account. However, the alternative of applying one common conversion factor, such as the weights used by Eurostat or those used in the Area-Wide model for the euro area (Fagan et al. 2001), does not change the results significantly. The alternative of conversion at current exchange rates is not appropriate, as it does not allow for difference in price levels among countries. Moreover, current exchange rates are volatile and affected by a number of factors, such as capital movements, trade flows, and the sentiment of financial markets, which makes them unsuitable to compare fluctuations in real economic activity across countries.

Appendix 2 Sectoral developments in individual euro area countries

The individual country results of the sectoral analysis are presented below (see Tables a2.1 and a2.2). In interpreting the data the following has to be borne in mind.

1. The differences in measured ALP in the ICT producing manufacturing sector are mainly due to differences in price deflators (see Table a2.3), which are probably mainly caused by methodological differences in the measurement of real output rather than by real economic differences. The use of hedonic deflators by France, for example, explains the relatively high productivity growth in the ICT producing manufacturing sector according to the French national data. The 'distortionary' effects of the use of different price deflators becomes apparent when a common deflator based on the United States is substituted for the national deflators of the euro area countries studied. These alternative calculations show high productivity growth in France comparable to those in the other euro area countries (see Table a2.4). Note in this context, that the national deflator in France is declining even more rapidly than that in the United States, implying that measured productivity growth is even slightly reduced in the case of France when using a common US-based price index.

2. The country tables show that in the course of the 1990s the share of ICT sectors only did increase in the case of Finland. The share has been more or less stable in the other euro area countries. Note that value added shares are calculated with current prices and thus are directly comparable across countries. The relatively high share of ICT producing sectors in value added is one of the factors explaining the faster overall productivity growth in Finland vis-à-vis the other euro area countries.

3. Moreover, the structure of the economy differs from country to country. For example, Finland has a large ICT producing manufacturing sector. In fact, it is even larger than that of the United States in relative terms. However, within the United States there are also large regional differences. In this context, it is interesting to note that Finland started relatively early with the liberalisation of its telecom market, as stressed by Colecchia and Schreyer (2001), which may be one of the reasons behind the rapid take-off of ICT production in Finland in the course of the 1990s. In this regard, it can also be noted that those countries that are commonly thought to have been among those that made most progress with labour market reforms in the euro area (namely Finland and the Netherlands), also show the largest overall employment growth.

Table a2.1 Share in value added

		Finland	France	Germany	Italy	Netherlands
	period					
ICT producing sectors, manufacturing	1991	0.6%	0.8%	1.3%	0.8%	1.6%
	1998	3.6%	0.8%	0.7%	0.5%	1.2%
ICT producing sectors, services	1991	3.2%	3.6%	3.3%	3.1%	2.8%
	1998	4.0%	3.8%	3.9%	3.7%	3.9%
ICT using sectors, manufacturing	1991	2.3%	3.3%	5.8%	3.2%	2.8%
	1998	3.0%	3.3%	4.5%	3.1%	2.3%
ICT using sectors, services	1991	6.5%	11.2%	10.1%	10.5%	11.1%
-	1998	6.0%	10.9%	11 1%	11.3%	13.4%

Source: own calculations using data from STAN OECD database, Groningen University ICT database, and Statistics Netherlands

Table a2.2 ALP growth

		Finland	France	Germany	Italy	Netherlands
	period					
ICT producing sectors, manufacturing	1991-1998	22.8%	24.0%	9.8%	-0.1%	5.9%
	1995-1998	21.1%	26.9%	11.8%	-1.9%	3.3%
ICT producing sectors, services	1991-1998	7.1%	3.7%	8.7%	5.3%	2.6%
	1995-1998	9.6%	5.8%	12.1%	5.0%	3.1%
ICT using sectors, manufacturing	1991-1998	5.5%	4.4%	3.8%	3.0%	5.8%
	1995-1998	4.5%	3.1%	2.5%	1.3%	1.0%
ICT using sectors, services	1991-1998	3.1%	-1.3%	0.5%	0.7%	0.7%
	1995-1998	3.0%	-1.6%	1.7%	-0.8%	0.1%

Source: own calculations using data from STAN OECD database, Groningen University ICT database, and Statistics Netherlands

Table a2.3 Change in implicit value added deflator

		Finland	France	Germany	Italy	Netherlands
	period			_	-	
ICT producing sectors, manufacturing	1991-1998	-2.4%	-15.4%	-3.6%	-0.5%	-2.0%
	1995-1998	-2.4%	-18.0%	-2.1%	0.3%	-1.6%
ICT producing sectors, services	1991-1998	-0.2%	-0.7%	-0.6%	2.3%	1.1%
	1995-1998	0.3%	-2.3%	-3.5%	3.1%	-2.4%
ICT using sectors, manufacturing	1991-1998	2.3%	0.5%	0.5%	2.2%	-1.4%
	1995-1998	0.3%	0.2%	0.1%	0.8%	-3.8%
ICT using sectors, services	1991-1998	1.7%	2.8%	1.2%	3.2%	3.5%
	1995-1998	1.2%	2.3%	-0.6%	3.2%	2.0%

Source: own calculations using data from STAN OECD database, Groningen University ICT database, and Statistics Netherlands

Table a2.4 ALP growth on the basis of US deflator

		Finland	France	Germany	Italy	Netherlands
	period					
ICT producing sectors, manufacturing	1991-1998	33.1%	16.4%	17.5%	10.4%	15.3%
	1995-1998	35.0%	18.8%	25.0%	12.3%	16.1%

Source: own calculations using data from STAN OECD database, Groningen University ICT database, and Statistics Netherlands

Appendix 3 Growth accounting for individual countries

The results of the growth accounting analysis for the individual euro area countries covered in this study are shown below, along with the results of other studies which made use of official data (see Tables a3.1 and a3.2). From Table a3.1, it can been seen that the contribution of ICT capital to ALP growth increased in all countries except Italy. Italy has also been lagging in terms of the ICT contribution at the sectoral level. Furthermore, all countries studied experienced a deceleration of TFP growth, with the exception of France. However, French TFP growth was already relatively low at the beginning of the 1990s. From Table a3.2 it emerges that in qualitative terms the estimates from this study and other related studies tend to yield a similar picture, especially on the increased importance of ICT for economic growth over the 1990s. However, the following caveats should be borne in mind in interpreting the quantitative estimates, which do differ somewhat from one study to the other.

1. This study focuses on growth rates of real GDP whereas most other studies tend to use real business sector output. The use of GDP in this study was motivated most importantly by the fact that the data on investment for three of the four countries are available for the total economy only and not for the private sector. In this study it was chosen to use these data directly, whereas for example Colecchia and Schreyer (2001) constructed their own estimates of business sector investment on the basis of the data on total economy investment in the cases of Germany and Italy, to enhance comparability with the United States. As the main focus of this study is to consider the importance of ICT for the euro area and not to establish results which are comparable with estimates for other economies, such an adjustment - which necessarily involves some arbitrary decisions and possibly creates (further) biases in the data – was considered unjustified in this case.

2. A comprehensive framework to calculate ICT contributions to output and productivity growth is used here. A comprehensive framework allows for a cross-check with the contribution of labour to output growth and for TFP estimates. The importance of such cross-checking emerges from a closer look at the study by Mairesse et al. (2000) for France. In particular, in this study the share of capital goods (excluding construction) adds up to 12.1% on average over the period 1995-1999. The total share of capital was 31.7% in that period according to the national accounts. The difference of 19.6 percentage points seems too much to be attributable to the exclusion of construction.¹⁴ The shares of capital assumed affect the estimates of the level of contributions of ICT capital to output growth (but not the pattern over the years). This probably leads to an

¹⁴ In our calculations the share of construction was 13.7%.

underestimate of contributions of ICT capital to output growth in the study of Mairesse et al.

3. The depreciation rate for software used in this study is higher than the depreciation rate used in some of the other studies. This is probably due to a different interpretation of the right concept of depreciation, an issue which is not straightforward to settle on theoretical grounds. The System of National Accounts (1993) defines depreciation as "the decline, during the course of the accounting period, in the current value of the stock of fixed assets owned and used by a producer as a result of physical deterioration, normal obsolescence, or normal accidental damage".¹⁵ It thus relates to economic depreciation as well as physical depreciation. In the presence of rapid technological change, high residual values of assets at the end of their assumed service lives are not consistent with a high rate of economic obsolescence. In particular, the net value of an intangible asset such as software can be expected to be close to zero at the end of its service life. This implies a relatively large declining balance rate and thus a relatively high depreciation rate. The depreciation rate used affect the size of the estimated contribution of software to output growth, with higher depreciation rates implying higher contributions. The depreciation rate for software used in this study allows for a depreciation to about 5% of the original value at the end of the service life, while the depreciation rate used in other studies tend to leave software at much higher residual values at the end of its service life (in the order of 25%). Consequently, other studies find lower contributions of software to output growth. The pattern of an increased contribution from the first to the second half of the 1990s is, however, not affected by this difference.

¹⁵ Note in this respect, that changes in the acquisition prices of assets, which are also part of the formula for user cost of capital, reflect revaluation – not depreciation. See Fraumeni (1997) for an alternative view. In particular, she argues that price changes reflect also obsolescence, which would overlap with the definition of depreciation from SNA.

Table a3.1 Decomposition of ALP growth
(percentages for ALP, percentage points otherwise)

		France	Germany	Italy	Netherlands
	period		-	-	
ALP	1991-1995	1.79	2.70	2.95	1.96
	1996-1999	1.37	1.53	0.86	0.53
Capital deepening					
ICT	1991-1995	0.26	0.20	0.19	0.34
	1996-1999	0.45	0.27	0.21	0.47
Information equipment	1991-1995	0.14	0.10	0.03	0.22
	1996-1999	0.18	0.12	0.03	0.26
Software	1991-1995	0.08	0.07	0.03	0.08
	1996-1999	0.21	0.13	0.09	0.20
Communications equipment	1991-1995	0.04	0.03	0.13	0.04
	1996-1999	0.06	0.01	0.09	0.01
Other capital	1991-1995	1.08	0.98	1.26	0.84
	1996-1999	0.39	0.62	0.47	-0.25
TFP	1991-1995	0.45	1.52	1.49	0.78
	1996-1999	0.52	0.64	0.18	0.30

Source: own calculations based on data from OECD and national accounts

France									
	Period	Contributions to output growth (percentage points) ¹							
		IT equipment	Software	Comm. equipment	TFP growth				
This study	1991-1995	0.14	0.07	0.03	0.45				
	1996-1999	0.19	0.23	0.08	0.52				
Colecchia/Schreyer	1990-1995	0.11 (incl. com. eq.)	0.02	NA	NA				
	1995-1999	0.19 (incl. com. eq.)	0.08	NA	NA				
Mairesse et al.	1989-1995	0.09	0.05	0.03	NA				
	1995-1999	0.13	0.09	0.05	NA				
German									
	Period	Contributions to output growth (percentage points) ¹							
		IT equipment	Software	Comm. equipment	TFP growth				
This study	1991-1995	0.09	0.06	0.02	1.52				
	1996-1999	0.12	0.13	0.01	0.64				
Colecchia/Schreyer	1990-1995	0.16 (incl. com. eq.)	0.06	NA	NA				
	1995-1999	0.14 (incl. com. eq.)	0.07	NA	NA				
Italy									
-	Period	Contributions to output growth (percentage points) ¹							
		IT equipment	Software	Comm. equipment	TFP growth				
This study	1991-1995	0.02	0.01	0.11	1.49				
	1996-1999	0.04	0.10	0.11	0.18				
Colecchia/Schreyer	1990-1995	0.10 (incl. com. eq.)	0.01	NA	NA				
	1995-1999	0.12 (incl. com. eq.)	0.04	NA	NA				
Netherland					2				
	Period	Contributions to labour p	roductivity g	growth (percentage p	points) ²				
		IT equipment	Software	Comm. equipment	TFP growth				
This study	1991-1995	0.22	0.08	0.04	0.78				
	1996-1999	0.26	0.20	0.01	0.30				
СРВ	1991-1995	0.2 (total ICT excl. s	oftware)		0.7				
	1996-1999	0.2 (total ICT excl. s	oftware)		1.3				
Van der Wiel	1991-1995	0.2 (incl. com. eq.)	0.1	NA	0.5				
	1996-2000	0.2 (incl. com. eq.)	0.2	NA	1.2				

Table a3.2 Comparison of this study with other studies on individual euro area countries

¹This study; GDP; other studies: business sector output

 $^2\mbox{This study; GDP}$ per hour worked; other studies: business sector output per employee fte

Appendix 4 Measurement problems related to productivity

The problem of measurement of macro-economic data has featured prominently in recent discussions on the impact of the new technologies on the economy. In particular, the techniques to account for quality changes of ICT production and expenditure in the national accounts have received particular attention. However, other measurement problems such as methods to take better account of changes in relative prices and those related to measuring services sector output may be as important in this regard. This appendix attempts to shed more light on these measurement issues.

4.1 The sensitivity of euro area real GDP to alternative deflators for ICT goods

As Wyckoff (1995) showed, the large differences in computer deflators between countries are at least partly due to methodological differences. In particular, conventionally constructed price indices would not fully capture quality improvements, as opposed to other indices such as those constructed applying the hedonic technique (i.e. a deflation technique based on a regression of the prices of a basket of goods on a set of qualities or characteristics of those goods, to identify price changes due to quality changes). Application of conventional price deflating techniques would thus lead to an overestimation of price changes and hence an underestimation of volume changes.¹⁶

As most euro area countries apply traditional techniques, it has sometimes been argued that measured real GDP growth in the euro area would be higher if prices of ICT goods were fully adjusted for quality improvements (e.g. Cecchetti 2000). The substitution of US price indices for ICT goods and services in the euro area national accounts has been advocated as a first step to enhance comparability between the growth rates recorded in the official statistics. Schreyer (2000) presents simulated effects on key economic variables (real output, private final consumption, government expenditure, investment, exports and imports) and productivity in five OECD countries, on the basis of such alternative (US based) price indices of ICT products. He concludes, however, that the impact of alternative price indices on real GDP growth tends to be small. Alternative (US based) deflators are applied below to the production side framework as presented in Section 3 of the main text.

¹⁶ Although Aizcorbe et al (2000) show that in theory the particular methodology of construction of price indices should not necessarily matter, in practice, countries applying a more conventional technique record smaller price declines in ICT goods than countries applying the hedonic technique.

Since the rate of growth of aggregate GDP or value added is a weighted average of the growth rates of value added by industry, it would appear straightforward to compute adjusted real value added measures by adjusting the constant-price value added for the relevant industries (for instance those producing ICT equipment), using alternative deflators. Multiplying the alternative growth rate of real value added thus obtained for the industries concerned by their share in total value added would yield an alternative economy-wide estimate of real value added (or real GDP) growth. This is not a valid approach, however.

National accounts guidelines recommend price and volume indices for value-added to be based on the so-called 'double-deflation' method, combining deflators of gross output and intermediate inputs. In the present context, this point is of importance as many industries consume intermediate ICT products whose price changes may be overstated as well. Thus, both output and input prices have to be adjusted to assess the full impact on measured value added and on total gross value added. If prices are adjusted for a product in one industry that is delivered to another industry, real value added in both industries is affected and the adjustment goes in opposite directions. Hence, full and internally consistent estimates of inter-industry effects on input and output price and volume adjustments, and their final impact on overall value added (or GDP) have to be assessed using detailed input-output tables.

The detailed sectoral gross value added data from the OECD Stan database were combined with information from 1995 input-output tables in order to construct series of gross output and inputs for a total of 24 sectors. The focus is here on the ICT producing manufacturing sectors, 'office accounting equipment and computer machinery industry' (30 ISIC) and 'radio, TV and communications equipment industry' (32 ISIC). For euro area countries the US deflator was substituted for the national one, after a correction for differences in domestic inflation with the United States for non-ICT hardware goods in the period from 1992 to 1998. This correction is deemed necessary in order to make the alternative deflator for ICT equipment independent of changes in inflation that prevail in the differences in domestic inflation for non-ICT hardware goods does not lead to significant differences in the results.

Chart a4.1 shows the implicit price deflator for ICT equipment for the United States along with both the original and the alternative deflators for the euro area. Between 1992 and 1998 the alternative (US based) deflator for ICT equipment in the euro area decreased by 10.8% on average, compared to a 4.1% average decline for the original deflator, based on national data.



Chart a4.1 Price deflators for ICT equipment - euro area and United States

Source: own calculations based on OECD Stan database and Eurostat 1995 input-output tables

Table a4.1 summarises the results. The table shows a decomposition of the total effect of the use of an alternative (US based) ICT deflator on measured real GDP growth in the euro area, distinguishing the effects of using an alternative deflator for ICT equipment <u>output</u> and of an alternative price index for the use of ICT equipment as <u>input</u>.

(percentages for real value added, percentage points direrwise)								
	1992	1993	1994	1995	1996	1997	1998	avg 1992-98
unadj.real value added growth	1.6	-0.7	2.2	2.1	1.3	2.0	2.6	1.6
adjustments due to ICT output adjustment	0.08	-0.02	0.02	0.26	0.21	0.18	0.37	0.16
ICT input adjustment	-0.03	0.03	-0.01	-0.17	-0.16	-0.13	-0.28	-0.11
total impact	0.05	0.01	0.01	0.09	0.06	0.05	0.09	0.05
adj.real value added growth	1.7	-0.7	2.2	2.2	1.4	2.0	2.7	1.6

Table a4.1 Impact on euro area real value added growth of using US based deflators for ICT equipment (percentages for real value added, percentage points otherwise)

Source: own calculations based on OECD Stan database, Groningen University ICT database, Statistics Netherlands, and Eurostat 1995 input output tables Note: figures may not add up due to rounding

Since the effects computed are typically small, the intermediate results are presented to two decimal points in the table, even though the margins of error would probably only allow rounding to at most one decimal point. The effects of the use of alternative deflators for ICT equipment for the 1992-1998 period averages 0.16 percentage point per year for gross output growth (which has an upward effect) and 0.11 percentage point for the growth rate of intermediate inputs (which has a negative impact on the growth rate of real value added). The combined effect of the gross output and intermediate input corrections on the growth rate of real gross value added (and hence on real GDP) thus cancels out to a large extent. Between 1992 and 1998 the net effect on real value added growth is only visible at the two decimal point level, averaging a mere 0.05 percentage point. Furthermore, the adjustments show no clear pattern over time, although adjustments tend to be larger in the second half of the 1990s than in the first.

Thus, the estimates from the production side suggest that the mechanical net impact of alternative deflators for ICT equipment on real GDP growth in the euro area is relatively small (albeit with some variation across years).

4.2 Other measurement issues

The issue of ICT deflators can not be discussed in isolation. For example, a consistent use of aggregation procedures to arrive at area-wide aggregates would be required as well. At present, several countries within the euro area use chain-type indices with annually changing weights in their national accounts to compute growth rates of real GDP and components, as is done in the United States, while other euro area countries do not. Chain-type indices use adjacent period weights to allow for changes in relative prices and output over time. By contrast, some euro area countries use a fixed weight basis. The difference between the two methods is small as long as relative weights do not change significantly over time. However, in the event of strong changes in the relative weights the use of a fixed basis leads to a distortion of the price and growth measurement, and this bias increases with the distance from the base period. According to EU standards for national accounts (ESA95), the use of annually chain-weighted measures is to be completed by 2005.

Apart from the issue of ICT deflators and aggregation procedures, there are other sources of measurement error which influence measured output and productivity. In particular, distinguishing between price and quantity components of output has become increasingly difficult as the share of services in total value added has increased over time. Identifying volume and price changes in services sectors is difficult for a number of reasons. First, there is a relative dearth of primary statistics for the service sector. Second, it is often conceptually more difficult to define the quantity of a particular service delivered than the quantity of a tangible good. In many cases output in services sectors is estimated on the basis of inputs, which implies a probable under-recording of productivity growth. As the size of the services sector has increased over time, it is likely that the measurement error has increased as well. For instance, in the euro area the share of services (including government) in nominal value added increased from 56.9% in 1980 to 69.4% in 1999. To what extent this will impact on the international comparability of data is difficult to determine. In principle, any measurement problems associated with the increasing share of services may differ across countries. Moreover, and more generally, different accounting practices among statistical agencies could lead to a dissimilar impact. For instance, there are substantial differences in methodologies for distributing ICT expenditure, software in particular, over final and intermediate uses (Lequiller 2001, Oulton 2001, Bundesbank 2001).

In summary, differences in existing statistical practices among countries appear to hamper comparisons of output and productivity across countries and across sectors. However, the biases will affect the aggregate output and productivity measures less. Thus, while measurement errors do cloud the picture, it seems unlikely that they are the major explanation for the lower recorded productivity growth in the euro area in recent years.

European Central Bank Working Paper Series

- I "A global hazard index for the world foreign exchange markets" by V. Brousseau and F. Scacciavillani, May 1999.
- 2 "What does the single monetary policy do? A SVAR benchmark for the European Central Bank" by C. Monticelli and O. Tristani, May 1999.
- 3 "Fiscal policy effectiveness and neutrality results in a non-Ricardian world" by C. Detken, May 1999.
- 4 "From the ERM to the euro: new evidence on economic and policy convergence among EU countries" by I. Angeloni and L. Dedola, May 1999.
- 5 "Core inflation: a review of some conceptual issues" by M. Wynne, May 1999.
- 6 "The demand for M3 in the euro area" by G. Coenen and J.-L. Vega, September 1999.
- 7 "A cross-country comparison of market structures in European banking" by O. de Bandt and E. P. Davis, September 1999.
- 8 "Inflation zone targeting" by A. Orphanides and V. Wieland, October 1999.
- 9 "Asymptotic confidence bands for the estimated autocovariance and autocorrelation functions of vector autoregressive models" by G. Coenen, January 2000.
- 10 "On the effectiveness of sterilized foreign exchange intervention" by R. Fatum, February 2000.
- II "Is the yield curve a useful information variable for the Eurosystem?" by J. M. Berk and P. van Bergeijk, February 2000.
- 12 "Indicator variables for optimal policy" by L. E. O. Svensson and M. Woodford, February 2000.
- 13 "Monetary policy with uncertain parameters" by U. Söderström, February 2000.
- 14 "Assessing nominal income rules for monetary policy with model and data uncertainty" by G. D. Rudebusch, February 2000.
- 15 "The quest for prosperity without inflation" by A. Orphanides, March 2000.
- 16 "Estimating the implied distribution of the future short term interest rate using the Longstaff-Schwartz model" by P. Hördahl, March 2000.
- 17 "Alternative measures of the NAIRU in the euro area: estimates and assessment" by S. Fabiani and R. Mestre, March 2000.
- 18 "House prices and the macroeconomy in Europe: Results from a structural VAR analysis" by M. lacoviello, April 2000.

- 19 "The euro and international capital markets" by C. Detken and P. Hartmann, April 2000.
- 20 "Convergence of fiscal policies in the euro area" by O. De Bandt and F. P. Mongelli, May 2000.
- 21 "Firm size and monetary policy transmission: evidence from German business survey data" by M. Ehrmann, May 2000.
- 22 "Regulating access to international large value payment systems" by C. Holthausen and T. Rønde, June 2000.
- 23 "Escaping Nash inflation" by In-Koo Cho and T. J. Sargent, June 2000.
- 24 "What horizon for price stability" by F. Smets, July 2000.
- 25 "Caution and conservatism in the making of monetary policy" by P. Schellekens, July 2000.
- 26 "Which kind of transparency? On the need for clarity in monetary policy-making" byB. Winkler, August 2000.
- 27 "This is what the US leading indicators lead" by M. Camacho and G. Perez-Quiros, August 2000.
- 28 "Learning, uncertainty and central bank activism in an economy with strategic interactions" by M. Ellison and N. Valla, August 2000.
- 29 "The sources of unemployment fluctuations: an empirical application to the Italian case" by S. Fabiani, A. Locarno, G. Oneto and P. Sestito, September 2000.
- 30 "A small estimated euro area model with rational expectations and nominal rigidities" by G. Coenen and V. Wieland, September 2000.
- 31 "The disappearing tax base: Is foreign direct investment eroding corporate income taxes?" by R. Gropp and K. Kostial, September 2000.
- 32 "Can indeterminacy explain the short-run non-neutrality of money?" by F. De Fiore, September 2000.
- 33 "The information content of M3 for future inflation" by C. Trecroci and J. L. Vega, October 2000.
- 34 "Capital market development, corporate governance and the credibility of exchange rate pegs" by O. Castrén and T. Takalo, October 2000.
- 35 "Systemic risk: A survey" by O. De Bandt and P. Hartmann, November 2000.
- 36 "Measuring core inflation in the euro area" by C. Morana, November 2000.
- 37 "Business fixed investment: Evidence of a financial accelerator in Europe" by P. Vermeulen, November 2000.

- 38 "The optimal inflation tax when taxes are costly to collect" by F. De Fiore, November 2000.
- 39 "A money demand system for euro area M3" by C. Brand and N. Cassola, November 2000.
- 40 "Financial structure and the interest rate channel of ECB monetary policy" by B. Mojon, November 2000.
- 41 "Why adopt transparency? The publication of central bank forecasts" by P. M. Geraats, January 2001.
- 42 "An area-wide model (AWM) for the euro area" by G. Fagan, J. Henry and R. Mestre, January 2001.
- 43 "Sources of economic renewal: from the traditional firm to the knowledge firm" by D. R. Palenzuela, February 2001.
- 44 "The supply and demand for eurosystem deposits The first 18 months" by U. Bindseil and F. Seitz, February 2001.
- 45 "Testing the Rank of the Hankel matrix: a statistical approach" by G. Camba-Mendez and G. Kapetanios, February 2001.
- 46 "A two-factor model of the German term structure of interest rates" by N. Cassola and J. B. Luís, February 2001.
- 47 "Deposit insurance and moral hazard: does the counterfactual matter?" by R. Gropp and J. Vesala, February 2001.
- 48 "Financial market integration in Europe: on the effects of EMU on stock markets" by M. Fratzscher, March 2001.
- 49 "Business cycle and monetary policy analysis in a structural sticky-price model of the euro area" by M. Casares, March 2001.
- 50 "Employment and productivity growth in service and manufacturing sectors in France, Germany and the US" by T. von Wachter, March 2001.
- 51 "The functional form of the demand for euro area MI" by L. Stracca, March 2001.
- 52 "Are the effects of monetary policy in the euro area greater in recessions than in booms?" by G. Peersman and F. Smets, March 2001.
- 53 "An evaluation of some measures of core inflation for the euro area" by J.-L. Vega and M. A. Wynne, April 2001.
- 54 "Assessment criteria for output gap estimates" by G. Camba-Méndez and D. R. Palenzuela, April 2001.

- 55 "Modelling the demand for loans to the private sector in the euro area" by A. Calza,G. Gartner and J. Sousa, April 2001.
- 56 "Stabilization policy in a two country model and the role of financial frictions" by E. Faia, April 2001.
- 57 "Model-based indicators of labour market rigidity" by S. Fabiani and D. Rodriguez-Palenzuela, April 2001.
- 58 "Business cycle asymmetries in stock returns: evidence from higher order moments and conditional densities" by G. Perez-Quiros and A. Timmermann, April 2001.
- 59 "Uncertain potential output: implications for monetary policy" by M. Ehrmann and F. Smets, April 2001.
- 60 "A multi-country trend indicator for euro area inflation: computation and properties" by E. Angelini, J. Henry and R. Mestre, April 2001.
- 61 "Diffusion index-based inflation forecasts for the euro area" by E. Angelini, J. Henry and R. Mestre, April 2001.
- 62 "Spectral based methods to identify common trends and common cycles" by G. C. Mendez and G. Kapetanios, April 2001.
- 63 "Does money lead inflation in the euro area?" by S. N. Altimari, May 2001.
- 64 "Exchange rate volatility and euro area imports" by R. Anderton and F. Skudelny, May 2001.
- 65 "A system approach for measuring the euro area NAIRU" by S. Fabiani and R. Mestre, May 2001.
- 66 "Can short-term foreign exchange volatility be predicted by the Global Hazard Index?" by V. Brousseau and F. Scacciavillani, June 2001.
- 67 "The daily market for funds in Europe: Has something changed with the EMU?" by G. P. Quiros and H. R. Mendizabal, June 2001.
- 68 "The performance of forecast-based monetary policy rules under model uncertainty" by A. Levin, V. Wieland and J. C.Williams, July 2001.
- 69 "The ECB monetary policy strategy and the money market" by V. Gaspar, G. Perez-Quiros and J. Sicilia, July 2001.
- 70 "Central Bank forecasts of liquidity factors: Quality, publication and the control of the overnight rate" by U. Bindseil, July 2001.
- 71 "Asset market linkages in crisis periods" by P. Hartmann, S. Straetmans and C. G. de Vries, July 2001.
- 72 "Bank concentration and retail interest rates" by S. Corvoisier and R. Gropp, July 2001.

- 73 "Interbank lending and monetary policy transmission evidence for Germany" by M. Ehrmann and A. Worms, July 2001.
- 74 "Interbank market integration under asymmetric information" by X. Freixas and C. Holthausen, August 2001.
- 75 "Value at risk models in finance" by S. Manganelli and R. F. Engle, August 2001.
- 76 "Rating agency actions and the pricing of debt and equity of European banks: What can we infer about private sector monitoring of bank soundness?" by R. Gropp and A. J. Richards, August 2001.
- 77 "Cyclically adjusted budget balances: An alternative approach" by C. Bouthevillain, P. Cour-Thimann, G. van den Dool, P. Hernández de Cos, G. Langenus, M. Mohr, S. Momigliano and M. Tujula, September 2001.
- 78 "Investment and monetary policy in the euro area" by B. Mojon, F. Smets and P. Vermeulen, September 2001.
- 79 "Does liquidity matter? Properties of a synthetic divisia monetary aggregate in the euro area" by L. Stracca, October 2001.
- 80 "The microstructure of the euro money market" by P. Hartmann, M. Manna and A. Manzanares, October 2001.
- 81 "What can changes in structural factors tell us about unemployment in Europe?" by J. Morgan and A. Mourougane, October 2001.
- 82 "Economic forecasting: some lessons from recent research" by D. Hendry and M. Clements, October 2001.
- 83 "Chi-squared tests of interval and density forecasts, and the Bank of England's fan charts" by K. F. Wallis, November 2001.
- 84 "Data uncertainty and the role of money as an information variable for monetary policy" by G. Coenen, A. Levin and V. Wieland, November 2001.
- 85 "Determinants of the euro real effective exchange rate: a BEER/PEER approach" by F. Maeso-Fernandez, C. Osbat and B. Schnatz, November 2001.
- 86 "Rational expectations and near rational alternatives: how best to form expecations" by M. Beeby, S. G. Hall and S. B. Henry, November 2001.
- 87 "Credit rationing, output gap and business cycles" by F. Boissay, November 2001.
- 88 "Why is it so difficult to beat the random walk forecast of exchange rates?" by L. Kilian and M. P. Taylor, November 2001.

- 89 "Monetary policy and fears of financial instability" by V. Brousseau and C. Detken, November 2001.
- 90 "Public pensions and growth" by S. Lambrecht, P. Michel and J.-P. Vidal, November 2001.
- 91 "The monetary transmission mechanism in the euro area: more evidence from VAR analysis" by G. Peersman and F. Smets, December 2001.
- 92 "A VAR description of the effects of monetary policy in the individual countries of the euro area" by B. Mojon and G. Peersman, December 2001.
- 93 "The monetary transmission mechanism at the euro-area level: issues and results using structural macroeconomic models" by P. McAdam and J. Morgan, December 2001.
- 94 "Monetary policy transmission in the euro area: what do aggregate and national structural models tell us?" by P. van Els, A. Locarno, J. Morgan and J.-P. Villetelle, December 2001.
- 95 "Some stylised facts on the euro area business cycle" by A.-M. Agresti and B. Mojon, December 2001.
- 96 "The reaction of bank lending to monetary policy measures in Germany" by A. Worms, December 2001.
- 97 "Asymmetries in bank lending behaviour. Austria during the 1990s" by S. Kaufmann, December 2001.
- 98 "The credit channel in the Netherlands: evidence from bank balance sheets" by L. De Haan, December 2001.
- 99 "Is there a bank lending channel of monetary policy in Spain?" by I. Hernando and J. Martínez-Pagés, December 2001.
- 100 "Transmission of monetary policy shocks in Finland: evidence from bank level data on loans" by J. Topi and J. Vilmunen, December 2001.
- 101 "Monetary policy and bank lending in France: are there asymmetries?" by C. Loupias,F. Savignac and P. Sevestre, December 2001.
- 102 "The bank lending channel of monetary policy: identification and estimation using Portuguese micro bank data" by L. Farinha and C. Robalo Marques, December 2001.
- 103 "Bank-specific characteristics and monetary policy transmission: the case of Italy" by L. Gambacorta, December 2001.
- 104 "Is there a bank lending channel of monetary policy in Greece? Evidence from bank level data" by S. N. Brissimis, N. C. Kamberoglou and G. T. Simigiannis, December 2001.
- 105 "Financial systems and the role of banks in monetary policy transmission in the euro area" by M. Ehrmann, L. Gambacorta, J. Martínez-Pagés, P. Sevestre and A. Worms, December 2001.

- 106 "Investment, the cost of capital, and monetary policy in the nineties in France: a panel data investigation" by J.-B. Chatelain and A. Tiomo, December 2001.
- 107 "The interest rate and credit channel in Belgium: an investigation with micro-level firm data" by P. Butzen, C. Fuss and P. Vermeulen, December 2001.
- 108 "Credit channel and investment behaviour in Austria: a micro-econometric approach" by M. Valderrama, December 2001.
- 109 "Monetary transmission in Germany: new perspectives on financial constraints and investment spending" by U. von Kalckreuth, December 2001.
- 110 "Does monetary policy have asymmetric effects? A look at the investment decisions of Italian firms" by E. Gaiotti and A. Generale, December 2001.
- III "Monetary transmission: empirical evidence from Luxembourg firm level data" by P. Lünnemann and T. Mathä, December 2001.
- 112 "Firm investment and monetary transmission in the euro area" by J.-B. Chatelain,A. Generale, I. Hernando, U. von Kalckreuth and P. Vermeulen, December 2001.
- 113 "Financial frictions and the monetary transmission mechanism: theory, evidence and policy implications" by C. Bean, J. Larsen and K. Nikolov, January 2002.
- 114 "Monetary transmission in the euro area: where do we stand?" by I. Angeloni, A. Kashyap, B. Mojon, D. Terlizzese, January 2002.
- 115 "Monetary policy rules, macroeconomic stability and inflation: a view from the trenches" by A. Orphanides, December 2001.
- 116 "Rent indices for housing in West Germany 1985 to 1998" by J. Hoffmann and C. Kurz., January 2002.
- 117 "Hedonic house prices without characteristics: the case of new multiunit housing" by O. Bover and P. Velilla, January 2002.
- 118 "Durable goods, price indexes and quality change: an application to automobile prices in Italy, 1988-1998" by G. M. Tomat, January 2002.
- 119 "Monetary policy and the stock market in the euro area" by N. Cassola and C. Morana, January 2002.
- 120 "Learning stability in economics with heterogenous agents" by S. Honkapohja and K. Mitra, January 2002.
- 121 "Natural rate doubts" by A. Beyer and R. E. A. Farmer, February 2002.
- 122 "New technologies and productivity growth in the euro area" by F. Vijselaar and R. Albers, February 2002.